Author’s response to reviews

Title: Enhanced caveolin-1 expression increases migration, anchorage-independent growth and invasion of endometrial adenocarcinoma cells

Authors:

   Natalia Díaz (nataliadiazvaldivia@gmail.com)
   Denisse Bravo (denbravorod@yahoo.com)
   Hernán Huerta (hernanhuertac@gmail.com)
   Soledad Henríquez (soledadhenriquez@yahoo.es)
   Fernando Gabler (gablerf@gmail.com)
   Margarita Vega (mvega@med.uchile.cl)
   Carmen Romero (cromero@hcuch.cl)
   Gareth Owen (gowen@bio.puc.cl)
   Claudia Calderón (ccalderon@med.uchile.cl)
   Lisette Leyton (lleyton@med.uchile.cl)
   Andrew F.G Quest (aquest@med.uchile.cl)

Version: 5
Date: 21 May 2015

Author’s response to reviews: see over
Krisha Mae Natan  
Journal Editor Office  
BMC Cancer

Dear Krisha

We are resubmitting to you our manuscript (MS: 2777858815881218) entitled: “Enhanced caveolin-1 expression favors migration, anchorage-independent growth and invasion of endometrial adenocarcinoma cells”, which now includes the additional editorial modifications requested. These are indicated in red in the manuscript and are detailed subsequently. We hope the manuscript is now acceptable for publication.

Yours Sincerely

Andrew Quest, PhD.

Center for Molecular Studies of the Cell (CEMC)  
Advanced Center for Chronic Diseases (ACCDiS)  
Facultad de Medicina  
Universidad de Chile  
Av. Independencia 1027  
Santiago, Chile  
Phone/fax: 56-2-7382015  
E-mail: aquest@med.uchile.cl
Díaz et al. (MS: 2777858815881218)
Point-by-point response to editorial requests

Comments
On the basis of the comments we have received we feel that your manuscript is suitable for publication in the journal. However, before we proceed with acceptance there are some further editorial revisions which we require you to make. Please note that your manuscript cannot be accepted for publication until these revisions have been satisfactorily addressed.

-------------------------------------------------------------------

Editor's Additional Request:

1.) Please include email address of all authors at the front of your manuscript file.

Response – we have included a list of all authors with their respective E-mails on the first page of the manuscript.

2.) Kindly moved funding information after "Acknowledgements" section.

Response – we have moved funding information to the section “Acknowledgements” as requested